**Author's response to reviews**

**Title:** Evaluation of wound healing activity of Allamanda cathartica L. and Laurus nobilis L. extracts on rats

**Authors:**

Shivananda B Nayak Dr (snayak@fms.uwi.tt)
Poorna Nalabothu Dr (poorikishore@yahoo.com)
Sandiford Sandiford (steve.sandiford@gmail.com)
Vidyasagar Bhogadi Dr (vsbhogadi@yahoo.com)
Andrew Adogwa Dr (adogwa01@yahoo.com)

**Version:** 7  **Date:** 31 March 2006

**Author’s response to reviews:** see over
The titles of the authors are removed and figures are cropped